Author’s response to reviews

Title: f-treeGC: A questionnaire-based family tree-creation software for genetic counseling and genome cohort studies

Authors:
Tomoharu Tokutomi (tktmshr@iwate-med.ac.jp)
Akimune Fukushima (akimunef@iwate-med.ac.jp)
Kayono Yamamoto (kyamamot@iwate-med.ac.jp)
Yasushi Bansho (bansho@holonic-systems.com)
Tsuyoshi Hachiya (thachiya@iwate-med.ac.jp)
Atsushi Shimizu (ashimizu@iwate-med.ac.jp)

Version: 3 Date: 08 Jun 2017

Author’s response to reviews:

Matteo Pasini, Ph.D.
Editor-in-Chief
BMC Medical Genetics
Jun 8, 2017

Subject: Submission of revised manuscript, reference ID MGTC-D-16-00303R3

Dear Dr. Pasini,

Thank you very much for your review of our manuscript entitled “f-treeGC: A questionnaire-based family tree-creation software system for use in genetic counseling and genome cohort studies” by T. Tokutomi et al. (ID MGTC-D-16-00303R3).
Please find enclosed our revised manuscript. We moved all tables at the end of the manuscript, leaving the reference to the position in the text, and included the titles and legends of the tables in ‘Titles and legends to figures and tables’.

We hope that you will find this revised manuscript satisfactory.

Yours sincerely,

Akimune Fukushima, MD., Ph.D.
Department of Clinical Genetics, School of Medicine, Iwate Medical University
19-1 Uchimaru, Morioka, Iwate 020-8505, Japan
Tel: +81-19-651-5111
Fax: +81-19-907-3711
E-mail: akimunef@iwate-med.ac.jp